\begin{table}%t3 
\caption{\label{tab:haan_source_properties}Sources from \citet{haan08}.}
%\centerline
 {\small \begin{tabular}{lccccccccccc}
\hline \hline
Object $^a$   && RA(2000)  & Dec(2000)  & $v_{\rm hel}$   & $D_{\rm L}$    & \ion{H}{i} line width  & CO line width & CO -Ref. & $f$ & $L_{\rm IR}$ & $\Sigma_{M}$ \\
& & [h] [m] [s] & [\degr] [\arcmin] [\arcsec] & [km~s$^{-1}$] & [Mpc] & [km~s$^{-1}$] & [km~s$^{-1}$] & & &[10$^{\rm 10}~L_{\odot}$] & [$M_{\sun}$~pc$^{-2}$]\\
\hline  
\object{NGC~3147} & & 10:16:53.65 & +73:24:02.7 & 2820 & 40.9 & 430 & 420    & 1 & 0.98 & 21.5 & 1.81\\
\object{NGC~3627} & & 11:20:15.03 & +12:59:29.6 & 727    & 6.6  & 431 & 409    & 2 & 0.95 & 86.6 & 2.82\\
\object{NGC~4569} & & 12:36:49.80 & +13:09:46.3 & --235   & 16.8 & 396 & 385    & 3       & 0.97 & 21.2 & 1.34\\
\object{NGC~4826} & & 12:56:43.69 & +21:40:57.5 & 408    & 4.1  & 372 & 373.75 & 4,5,6,7 & 1.00 & 86.8 & 0.43\\
\object{NGC~5953} & & 15:34:32.39 & +15:11:37.7 & 1965 & 36   & 349 & 230    & 1       & 0.66 & 25.0 & 0.94\\
\object{NGC~6574} & & 18:11:51.23 & +14:58:54.4 & 2282 & 38   & 340 & 350    & 1       & 1.03 & 35.7 & 1.88\\
\object{NGC~6951} & & 20:37:14.09 & +66:06:20.3 & 1424 & 24.1 & 338 & 380.2  & 8       & 1.12 & 34.8 & 1.71\\
\object{NGC~1961} & & 05:42:04.80 & +69:22:43.3 & 3934 & 53.9 & 767 & 731    & 9       & 0.95 & 20.1 & 12.84\\
\object{NGC~3368} & & 10:46:45.74 & +11:49:11.8 & 897    & 8.1  & 399 & 365    & 10      & 0.91 & 24.4 & 1.33\\
\object{NGC~3718} & & 11:32:34.85 & +53:04:04.5 & 994    & 17.0 & 465 & 470    & 11      & 1.01 & 5.38 & 1.74\\
\object{NGC~4321} & & 12:22:54.90 & +15:49:20.6 & 1571 & 16.8 & 291 & 200$^c$ & 12 & 0.69 & 50.5 & 2.91\\
\object{NGC~4579} & & 12:37:43.52 & +11:49:05.5 & 1519 & 37   & 374 & 380    & 13      & 1.02 & 15.1 & 1.26\\
\object{NGC~4736} & & 12:50:53.06 & +41:07:13.7 & 308    & 4.3  & 284 & 270    & 5  & 0.95 & 126  & 1.53\\
\object{NGC~7217} & & 22:07:52.38 & +31:21:33.4 & 952 & 16.0 & 326 & 320    & 14 & 0.98 & 15.8 & 2.13\\
\object{NGC~2782} & & 09:14:05.11 & +40:06:49.2 & 2562 & 37.3 & 246 & 260    & 1 & 1.06 & 20.8 & 3.29\\
\object{NGC~5248} & & 13:37:32.07 & +08:53:06.2 & 1153 & 15   & 311 & 300    & 5 & 0.96 & 45.7 & 2.33\\
\hline
\end{tabular}}
\medskip
$^a$ Taken from \citet{haan08}; $^b$~references: 1 --~\citet{you95};
2 --~\citet{sage93}; 3 --~\citet{bos95}; 4 --~\citet{helf03};
5 --~\citet{nishi01}; 6 --~\citet{cas93}; 7 --~\citet{aal91};
8 --~\citet{koh99}; 9 --~\citet{tut97}; 10 --~\citet{brai93};
11 --~\citet{pott04}; 12 --~\citet{saka95}; 13 --~\citet{sof03}; 14 --~\citet{com04}; $^c$~uncertain.
\vspace*{5mm}
\end{table}